Author's response to reviews

Title: A randomized controlled study of a social skills training for preadolescent children with autism spectrum disorders: Generalization of skills by training parents and teachers?

Authors:

Vera Dekker (v.dekker@accare.nl)
Maaike H Nauta (m.h.nauta@rug.nl)
Erik J Mulder (e.j.mulder@accare.nl)
Marieke E Timmerman (m.e.timmerman@rug.nl)
Annelies De Bildt (a.de.bildt@accare.nl)

Version: 2 Date: 16 June 2014

Author's response to reviews: see over
Groningen, June 16th, 2014

Concerns: resubmission of manuscript version 2 to BMC Psychiatry

Dear Editor,

On Tuesday 27 May we have submitted our manuscript entitled “A randomized controlled study of a social skills training for preadolescent children with autism spectrum disorders: Generalization of skills by training parents and teachers?”.

You asked us to make some changes in our manuscript. Below a summary of the changes we made according to your request:

- In the manuscript we have now specifically stated that the authors declare that they have no competing interests.
- We have forwarded the requested documentation about the ethical and funding approval to BMCSeriesEditorial@biomedcentral.com.

We hope that everything is complete now and that you will consider publishing our design paper in BMC Psychiatry.

Yours sincerely,

Also on behalf of the other authors,

Awaiting your editorial decision,

Vera Dekker

Vera Dekker, M.Sc.
PhD student
Accare, Child and Adolescent Psychiatry
University Medical Center Groningen
PO Box 660
9700 AR Groningen
The Netherlands
E: v.dekker@accare.nl
T: +31.50.3681147